Author’s response to reviews

Title: The anti-apoptotic, antioxidant and anti-inflammatory effects of curcumin on acrylamide-induced neurotoxicity in rats

Authors:

Jie Guo (1434550359@qq.com)
Xiaolu Cao (534427581@qq.com)
Xianmin Hu (huxianmin@wust.edu.cn)
Shulan Li (2283134613@qq.com)
Jun Wang (wangjun@wust.edu.cn)

Version: 2 Date: 03 Aug 2020

Author’s response to reviews:

Dear Editor:

We have to apologize for some textual overlap in our paper with other previously published works. As your kindly reminder, there is some overlap with the article https://journals.sagepub.com/doi/10.1177/1934578X20910548, the latter of which is the other study conducted by our team. In the revised manuscript, we have avoided the textual overlap with this article.

In addition, we amazedly found that our results were included in the web http://www.turmeric-curcumin.com/ according to your kindly reminder. This web seemed to get our results from https://www.researchsquare.com/article/rs-7396/v1, which has been show in the web http://www.turmeric-curcumin.com/. And we emailed to the support@turmeric-curcumin.com, and required the supporters of this web to delete our results from this web. Unfortunately, we haven't heard back from the supporters yet. In order to avoid the overlap in the "Results" section with http://www.turmeric-curcumin.com/, we revised the "Results" section in the ABSTRACT in the revised manuscript.

We hope that the manuscript can fit the quality of the journal this time. Thank you very much for your consideration!

Yours sincerely, Jun WANG